Author's response to reviews

Title: Effect of octanoic acid-rich formula on plasma ghrelin levels in cachectic patients with chronic respiratory disease

Authors:

Jun-ichi Ashitani (jashi2@fc.miyazaki-u.ac.jp)
Nobuhiro Matsumoto (nobu@fc.miyazaki-u.ac.jp)
Masamitsu Nakazato (nakazato@fc.miyazaki-u.ac.jp)

Version: 2 Date: 15 May 2009

Author's response to reviews:

Dr. Nehme Gabriel,
Editor-in-Chief, Nutrition Journal

May 15, 2009

Dear Editor,

We are deeply grateful for your evaluation. Enclosed are the revised manuscript MS: 1012417642260937 entitled “Effect of octanoic acid-rich formula on plasma ghrelin levels in cachectic patients with chronic respiratory disease”. We have considerably revised our manuscript by responding to the questions and concerns of the reviewers. We would appreciate it if you would accept our manuscript.

Sincerely yours,
Jun-ichi Ashitani, MD.
Department of Internal Medicine,
Medical University of Miyazaki,
5200 Kihara, Kiyotake, Miyazaki, 889-1692, Japan
Tel: +81-985-85-2965
Fax: +81-985-85-1869